RESEARCH ARTICLE

Open Access

How can we recruit more men of African or African-Caribbean ancestry into our research? Co-creating a video to raise awareness of prostate cancer risk and the PROFILE study

Emma Hainsworth^{1*}, Eva McGrowder², Jana McHugh², Elizabeth Bancroft², Sean Mahabir¹, Winston Webber³, Rosalind Eeles² and Susanne Cruickshank^{1,4}

Abstract

Background: Men of African ancestry are at increased risk of developing prostate cancer (PrCa) compared to men from other backgrounds. The PROFILE study aims to understand whether genetic information can better target who needs PrCa screening. PROFILE has so far had difficulty reaching men of African or African -Caribbean ancestry to take part. In this involvement project we worked in partnership with a group of such men to co-create a video to raise awareness of PrCa risk amongst this community and promote participation in the study.

Methods: We recruited seven men of African or African-Caribbean ancestry who completed an initial survey on the Cancer Patients'Voice platform. We then held an online discussion panel and maintained contact to encourage dialogue and planning of the video. Utilising a participatory approach, the ideas for the video were decided in collaboration with the panel who held expert knowledge of various communities and understood the messages that would best resonate and engage with other men of the same origins. Once the video had been edited and finalised, two members of the group expressed interest in writing up the project and are listed as co-authors.

Results: The video in its entirety was driven by the panel's ideas. The choice of a barber shop setting; leading with a positive case study and highlighting the importance of men's family members rather than a focus on scientific language, statistics or researchers were all features that were discussed and agreed upon by the panel. The men shared the video within their networks. It was placed on websites and promoted as part of a social media campaign during Black History Month.

Conclusions: Groups with the greater healthcare needs and the most to gain from advances in care and treatment can often be the most excluded from research participation. This is pertinent in PrCa research where men of African or African-Caribbean ancestry are at greater risk. The project gave equal power and decision making to the men and provides an example of successful inclusive involvement. The result was a unique approach to making a study video.

Keywords: Patient and public involvement, Inclusion, Diversity, Prostate cancer, Co-creation

Full list of author information is available at the end of the article



^{*}Correspondence: emma.hainsworth@rmh.nhs.uk

¹ Royal Marsden NHS Foundation Trust, 203 Fulham Rd, London SW3 6JJ,

Plain English summary

This project involved working in partnership with men of African or African-Caribbean ancestry to co-create a video intended to raise awareness of prostate cancer (PrCa) risk and promote participation in a genetic screening study called PROFILE. Men of African or African-Caribbean ancestry are at increased risk of developing PrCa compared to other men. The PROFILE study aims to understand whether genetic information can better target who needs PrCa screening. The study has had problems recruiting men from these communities. Methods: We engaged seven men of African or African-Caribbean ancestry: three PROFILE study participants and four from the Race, Ethnicity and Cultural Heritage (REACH) staff forums across the Royal Marsden Hospital and the Institute of Cancer Research. They completed a survey, joined an online discussion panel and we continued working together. The group decided on the structure and content of the video; to include a PrCa survivor who had been successfully screened and treated early for his disease, and a daughter of one of the panellists. The men were also involved in the dissemination plans of the finished video, and two agreed to be co-authors of this paper. Findings: Features of the video led by the men included the choice of a barber shop setting; leading with a positive shared story and highlighting the importance of family rather than science, statistics or researchers. Dissemination: The group shared the video within their networks. It was placed on websites and promoted as part of a social media campaign during Black History Month.

Background

This paper reports a patient and public involvement project conducted in partnership with men of African or African-Caribbean ancestry to plan, create and disseminate a video to raise awareness of prostate cancer risk and promote participation in a genetic prostate cancer screening study, the PROFILE study (NCT02543905). For clarity in terminology, we refer herein to men of African or African-Caribbean ancestry as men of black African or black African-Caribbean ancestry—which is defined within the study protocol as having both parents and all four grandparents of that origin.

The PROFILE study

Prostate cancer is the most common cancer in men in the United Kingdom (UK) with men of African or African-Caribbean ancestry at 2–3 times the risk of developing it in comparison to their counterparts of European ancestry, and with a 30% higher mortality rate [1]. We do not understand enough about why some men, including men of African or African-Caribbean ancestry and men of European ancestry with a family history of the disease are at higher risk. The PROFILE study aims to address this gap in the knowledge by looking at the genes of healthy men at higher risk including men of African or African-Caribbean ancestry. The PROFILE study monitors for signs of prostate cancer using blood tests, and if required, scans and biopsies are undertaken, and participants are followed up over 5 years. The aim of the research is to understand whether genetic information can be used as a screening tool to target those at higher risk of developing prostate cancer. To date, the PROFILE study has experienced some difficulty in recruiting men of African or African-Caribbean ancestry and we wanted our involvement work to address this. Evidence in the literature supports using involvement activity to improve study recruitment, particularly if this involvement includes working in partnership with people who have relevant lived experience [2, 3].

Inclusive involvement

In this project we wanted to work in partnership with a group of men of African or African-Caribbean ancestry and share learning so that we could create a video that would best engage these communities. In doing so and writing up our experience we hope to add to the limited evidence that exists on inclusive involvement.

We know that the groups with the greater healthcare needs and the most to gain can be the most excluded from healthcare decision-making. Covid-19 has made everyone more aware of health inequalities and has starkly highlighted the unacceptable situation of poorer health outcomes for patients from Asian and Black backgrounds [4]. There is evidence that these groups are less likely to have participated in research [5] and that more widely, the UK health system often mirrors the forces that undermine the health of people from ethnic minority backgrounds [6]. There are numerous barriers to ethnic minority participation in research which are complex, and these include structural issues ranging from lack of access to care, distrust in healthcare systems, to discrimination. It is therefore crucial to report and describe involvement which explicitly attempts to address power imbalances and reach groups that have been underserved in such a way so that evidence can be built, and lessons can be shared to improve health equity.

There is a current drive encouraged by major national funders to offer public involvement opportunities that are accessible and reach people and groups according to research need. Despite being an ambition, it is not clear that this is happening to any great extent. There is a considerable gap in the literature with only a very small number of health and social care studies outside of the United States (US) including any African or African-Caribbean and other minority ethnic involvement. Those studies that have written about them, showed that involvement was limited to the early stages of the research process, with poor reporting on the details of facilitators or barriers [7].

Power relations

Public involvement in research means research that is done 'with' or 'by' the public, not 'to', 'about' or 'for' them [8]. Despite this aspiration of a more equal partnership, current models of Patient and Public Involvement (PPI) can often reproduce imbalances of power within health-care where patients/public can find themselves invited into spaces which give them little scope to do anything other than legitimise decisions which have been made by others [9, 10]. This can be seen in professionally dominated approaches with a business meeting or committee format which maintains the appearance of inclusion but has the effect of excluding those who may not feel comfortable in this setting. We wanted to conduct our involvement activity with a more equal power balance between researchers and those involved.

Measuring impact

The increased focus on measuring the impact of involvement activity can have the effect of applying scientific, rigid notions of the methods of measurement to a process which can be more a social collaboration with learning between researchers and the public as a valuable end in itself [11]. The impact of involvement is not limited to the experience of the patient or public participants and the effect on the research project, researchers themselves also learn new knowledge through involvement which can change their priorities and attitudes [12, 13]. Our project aimed to offer an inclusive involvement opportunity that would follow the approach of participatory research by encouraging the development of equal power relationships and shared learning. In our reporting we have adopted a qualitative narrative rather than the use of a standardised checklist as we feel that this better reflects the social collaborative partnership of the work [14]. We ensured that involvement continued throughout the project by including two men from the group as co-authors in our publication.

Methods

Aims and objectives

Our objective was to work in partnership with a group of men of African or African-Caribbean ancestry to plan and co-create a video. This video had two aims: to raise awareness about prostate cancer risk in men from these communities, and to encourage participation in an existing genetic screening study, PROFILE.

Recruitment into discussion panel

We held a discussion panel of seven men of African or African-Caribbean ancestry in an online meeting via ZOOM[™] Video Communication in December 2020. The panelists came from a mix of African and African-Caribbean backgrounds and a range of ages. Three were existing PROFILE study participants who had been approached by members of the clinical team prior to speaking to the authors; four others were members of what was then known as the Black, Asian and Minority Ethnic (BAME) staff forums, now called the Race, Ethnicity and Cultural Heritage forums (REACH). One further member of this forum was not able to attend the online meeting. His work role meant that he did not have access to a computer or the necessary free time to attend the discussion. In this instance one of the authors met him in person during one of his breaks. All members of the group were reimbursed for their time according to the National Institute for Health Research (NIHR) National Standards for Public Involvement [15]. In the discussion they shared their ideas about what representation and messages they thought should be included in the proposed video to engage men within the African or African-Caribbean communities.

Participation in the video

In the discussion panel, men agreed that the video should feature a 'success story' of someone who had been diagnosed early with prostate cancer due to screening. We used our clinical network to help approach a man whose prostate cancer had been detected early and successfully treated to share his story within the video. It was also felt to be important to include a loved one or family member to share their perspective on the importance of their relative being aware of prostate cancer risk and screening, so we also included a daughter and her father, who was taking part in the PROFILE study. We included two more PROFILE study participants from our discussion panelists, and this made a total of five participants who we agreed were to be filmed within the video. The group was given the opportunity to review and comment upon the draft edit before it was finalised.

Writing up the project for publication

We wanted the collaborative nature of this project to extend beyond the making of the video and into the dissemination stage. Some of the group members were active within their own community networks and were encouraged to share the finished video across these networks. The opportunity to co-author this paper was advertised to all who were involved in this involvement project, and two members expressed their interest to take part in the writing process. We held an online meeting lasting one hour, where they agreed to be co-authors, and in which they discussed priorities for what should be included. We then communicated by email in the exchange of draft text for comment and approval. The focus of their contribution was their views on the important learning from the project and how the video had been shaped by the involvement of the group. They were also happy to be included as named authors.

Results

How was the video shaped by the group's input and ideas? Leading with a positive case study

The group agreed that positive stories would be the most impactful and there was widespread support for starting the video with a case study of a man who had been screened early, treated for prostate cancer, and was now living well. They thought that this would provide a challenge to commonly held beliefs within communities that cancer was 'a death sentence' and could help to dispel fears. The video ended with the same man stating: 'You can survive cancer; I have survived cancer.'

The importance of family members

At the planning stage the group thought it would be important to include family members' perspectives on men looking after their health as this would resonate with their own relationships and would provide personal motivation. Talking about their personal prostate cancer risk or seeking help for symptoms could then be seen within a context of being strong and caring for your loved ones and making sure that they do not experience worry. As the prostate is linked to a masculine identity and virility, it was thought important to frame the topic in a way which did not threaten this identity. The group discussed the way in which women play their part in keeping healthy for their family by having breast and cervical screening; therefore, men being aware of prostate cancer risk and seeking help could be seen as men doing the same, and 'playing their part'. This is another example of the positive and proactive framing that the group believed would best engage the audience. Reflecting on the finished video, two of our patient/public authors agreed that the daughter encouraging her father to become aware of prostate cancer risk and to be checked out for any symptoms in the same way that women do routinely, conveyed a powerful message:

'Support from family, from wives, partners, children, I think that's really good to egg men on to get the checks done' (SM, public co-author).

WW thought that this message might also inspire female viewers to speak to their male relatives:

'A wife could watch it and think of her husband having symptoms, or a daughter might think of her father, and they might now feel confident to say something' (WW, patient co-author).

Prostate cancer affects younger black men too

Men who appeared in the video were of different ages to reflect the inclusion criteria of the PROFILE study (40 to 69 years). The inclusion of younger men in the video was designed to challenge a commonly held perception that prostate cancer or cancer in general is only a problem for an older age group. WW felt that their participation in the study role-modelled proactive health behaviour which was not common amongst his own contemporaries:

'Here are men who have not yet been diagnosed, willing to be monitored, they're willingly and joyfully adhering to it.....That's a message that so far I've found difficult to pass on to my friends' (WW, patient co-author).

SM's participation in the project as a member of the planning group and a co-author had an influence on his own behaviour, so much so that he requested a PSA test as part of a routine health-check as a result; something he would otherwise not have thought to request.

No scientific language, statistics, or researchers

The men unanimously agreed that we should avoid scientific language, statistics, death rates and tragedies as this would immediately cause people to switch off. They thought that researchers or the clinical team should not play a prominent role in the video and that since the messaging was aimed at healthy individuals, that the setting should be neutral and away from a clinical background. This was a surprise to the researchers and challenged their preconceived ideas about what the video might look like and stood in striking contrast to many of the cancer awareness and study promotion videos in circulation, which included just that sort of clinical setting and style of information.

The barber shop and its emotional resonance

On camera, men spoke with warmth and feeling about the importance of a barber shop in their lives. The location of the video had been unanimously agreed by the group as a setting which represented a social meeting place, where sensitive subjects could be discussed safely and freely, and where men got away from the pressures of everyday life and felt better about themselves. Talking about something as personal and sensitive as prostate cancer risk felt easier in this venue:

'You are just free to talk...it allowed us to be so natural. We're not doing a clinical interview; it's just a barber shop' (WW, patient co-author).

The video's impact on recruitment into the PROFILE study

The PROFILE study recruits men for targeted prostate cancer screening in two cohorts: men of African or African- Caribbean Ancestry and men of European Ancestry with a family history of prostate cancer. Initially, the PROFILE study opened in 2009 as a pilot study (Integrated Research Application System (IRAS): 7951; Research Ethics Committee (REC) reference: 09/ H0801/19) with just the European ancestry family history cohort. In 2015, the PROFILE full study opened with two cohorts: men of African or African- Caribbean Ancestry and men of European Ancestry with a family history of prostate cancer (IRAS: 132999; REC: 13/LO/1787; NCT02543905). Table 1 shows recruitment figures in both cohorts since the study began in 2015, with the family history cohort recruitment remaining relatively constant over the years. However, recruitment to the African or African-Caribbean cohort was poor, with no recruitment at all to this cohort during the first year of this study (2015) when compared to 47 participants being recruited to the family history cohort in the same year. The first participant to the African or African-Caribbean cohort was recruited in 2016, and this being the only recruit to this cohort until 2018, when compared to 96 participants recruited to the family history cohort during the period

Table 1 PROFILE recruitment figures

Recruitment year	Men of African and African-Caribbean ancestry recruited	Men of European ancestry with Family history of prostate cancer recruited
2015	0	47
2016	1	69
2017	0	27
2018	10	53
2019	32	47
2021	103	45

of 2016-2018 (Table 1). From 2018, initiatives were taken by the study team to improve recruitment to the African or African-Caribbean cohort, including the introduction of the study to Primary Care recruitment with multiple GP practices; study posters; engagement efforts within this community and recently, the PPI engagement video to further raise awareness of prostate cancer risk in this population group. The video was made available online and was promoted from August 2021 alongside a wider initiative with a mail-out conducted by GP practices highlighting the health risk to black men. The recruitment figures in the African or African-Caribbean cohort have increased three-fold in 2021 compared to previous years. Although it is not possible to attribute all of this increase to the video, this trend in recruitment is encouraging and does suggest a positive impact.

Discussion

This was an involvement activity which set out to centre the voices of men of African or African-Caribbean ancestry and facilitate the delivery of their ideas for raising awareness and engagement around prostate cancer risk. The PROFILE study had previously tried several approaches to raise awareness in the black community without significant success. Some of these included engaging community leaders such as church leaders, counsellors, community volunteers, general practitioners; displaying study posters in local businesses, churches, leisure centres; attending community meetings and holding awareness talks. All these approaches were well received by the public; however, they did not fully translate in men expressing interest in the PROFILE study or to seek further information about prostate cancer. What was striking from these public talks was that it was mainly women who were keen to seek more information about prostate cancer, the types of symptoms to look out for and generally, what the PROFILE study was about. They would take study material, participant information sheets to pass on to their male family members, with the hope that they would then contact the study team and or take part in the study. Very few men expressed interest in the study through these approaches. The issue was not the lack of identifying participants, but rather failure to engage them fully. We therefore had to change our strategy and focus on raising awareness in these communities.

A recent NIHR public contributors' feedback survey of 2019 revealed that there was a lack of diversity in the people who took part in involvement activity, with younger people and minority ethnic groups under-represented [16]. This gap is clear within the literature. Describing this work of inclusive involvement from the conception of the project through to its dissemination is an important addition. By writing up this activity descriptively,

it is hoped that this will usefully contribute to existing evidence which can inform and guide future inclusive projects.

Cultural views and beliefs can affect the way men of African or African-Caribbean ancestry engage with information about cancer risk and screening, and health information based solely on the clinical meaning of cancer can be incongruent [17]. By seeking the group's views on what features of a video would resonate culturally with men of African or African-Caribbean ancestry, we hoped to engage these communities more effectively. The barber shop setting was an example of this, presenting the researchers with a novel idea that challenged their preconceptions and led them to venture outside of the comfort zone of the biomedical setting to find somewhere in the community that fitted the brief. The setting was remarked upon as one of the most engaging aspects of the video and testament to the need for researchers to be led in directions that did not feel familiar or immediately comfortable.

This project was funded by an NIHR Biomedical Research Centre (BRC) award dedicated to support a discrete Patient Public Involvement and Engagement activity. As such, it could accommodate the time required to build and sustain relationships; time which is not always available or well-aligned to the more usual research funding timetables [18]. As the planning work was held at a time of Covid-19 restrictions, the discussions had to take place online. In some senses this made for more equitable participation as it did not require travel or attendance in person. In the case of one member of the group whose work did not accommodate the use of a computer, one of the authors met him separately in his place of work. This prioritisation of relationships over the usual structure of formal meetings worked well to address power differentials which can stifle involvement [19] and in this case facilitated contributions.

Limitations

There is a large range of cultural and ethnic diversity within the African and African-Caribbean population in the UK and the seven men who participated in this project, although from a variety of backgrounds, could not represent this diversity in its entirety. These men were members of the staff REACH forums, study participants and a cancer survivor, and as such, were likely to be positive about medicine and research. Although they had an understanding of the issues, they did not represent the voices of men living in these communities who may have feelings of disenfranchisement and mistrust of the healthcare system in the UK.

Conclusions

This project provides an example of a successful inclusive involvement activity which has been written up descriptively to share good practice and learning for future work trying to increase diversity within study recruitment. The video created by the involvement group contained features and messages designed to resonate culturally with men of African or African-Caribbean ancestry and move beyond the provision of purely clinical information. Both the final video and the process of its creation aimed to engage and build trust to ensure that the study could better recruit and benefit those men most affected by prostate cancer.

Abbreviations

BAME: Black, Asian and Minority Ethnic; BRC: Biomedical Research Centre; IRAS: Integrated Research Application System; NHS: National Health Service; NIHR: National Institute for Health Research; PPI: Patient and Public Involvement; PrCa: Prostate Cancer; PSA: Prostate-Specific Antigen; REACH: Race, Ethnicity and Cultural Heritage; REC: Research Ethics Committee; UK: United Kingdom; US: United States.

Acknowledgements

We would like to thank all those who took part in our discussion panel and in the video itself including Frederick Forster, Olivier Hurtis, Winston Webber, Richard Borges-Da-Silva, Anne-Marie Borges-Da-Silva, Michael Ranes, Sean Mahabir, Georges Konin, Rudyard Lewis and Denzil James. We also wish to thank the producers of this video, Short Form Film Company, as well as Champs Barbers who so generously allowed us to film in their barber shop.

Author contributions

EH and EM led the involvement project with support from the PROFILE clinical team (RE, JM, EB) and SC. EH and EM were major contributors in writing the manuscript. SM and WW contributed to the writing as public and patient authors. All authors read and approved the final manuscript.

Funding

The video and involvement project was funded by a NIHR Biomedical Research Centre at The Royal Marsden and Institute of Cancer Research Award. The PROFILE study is funded by PCUK Movember Centre of Excellence, Cancer Research UK, The Ronald and Rita McAulay Foundation and The NIHR Biomedical Research Centre at the Royal Marsden and Institute of Cancer Research.

Availability of data and materials

The full length video can be accessed via YouTube $^{\text{TM}}$ using the following link: https://www.youtube.com/watch?v=A72hCjLN3gE

Declarations

Ethics approval and consent to participate

The activities outlined in this publication classify as community engagement and involvement with research as opposed to being research. Therefore, ethics approval was not required. The PROFILE study was reviewed by the London-Chelsea Research Ethics Committee under reference number: 13/LO/1787 and was approved to be conducted in National Health Service (NHS) England by the Health Research Authority under reference number: IRAS132999. These approvals included activities to raise awareness. The PROFILE study participants who took part were approached by members of the clinical team and gave verbal or email consent to being approached by the authors and receiving further information about the involvement project. For members of the REACH staff forums at the Royal Marsden Hospital and the Institute of Cancer Research, the opportunity was advertised and circulated by internal communication and those who were interested contacted the authors. Video participants provided their written consent on a Royal Marsden

NHS Foundation Trust media consent form on the day and filming took place with all the required Covid-19 safety precautions and social distancing was observed. All video participants received further reimbursement for their time and travel to the filming location according to the NIHR National Standards for Public Involvement [15].

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Royal Marsden NHS Foundation Trust, 203 Fulham Rd, London SW3 6JJ, UK. ²The Institute of Cancer Research SRD Building, 1st Floor D1S5, 15 Cotswold Road, Sutton SM2 5NG, Surrey, UK. ³Patient Author, London, UK. ⁴Faculty of Health Sciences and Sport, University of Stirling, Stirling, UK.

Received: 21 December 2021 Accepted: 5 April 2022 Published online: 18 April 2022

References

- Jones AL, Chinegwundoh F. Update on prostate cancer in black men within the UK. Ecancermedicalscience. 2014;8:455. https://doi.org/10. 3332/ecancer.2014.455.
- Crocker JC, Ricci-Cabello I, Parker A, Hirst JA, Chant A, Petit-Zeman S, et al. Impact of patient and public involvement on enrolment and retention in clinical trials: systematic review and meta-analysis. BMJ. 2018;363: k4738. https://doi.org/10.1136/bmj.k4738.
- Vogsen M, Geneser S, Rasmussen ML, et al. Learning from patient involvement in a clinical study analyzing PET/CT in women with advanced breast cancer. Res Involv Engagem. 2020;6:1. https://doi.org/10.1186/ s40900-019-0174-v.
- Apea VJ, Wan YI, Dhairyawan R, et al Ethnicity and outcomes in patients hospitalised with COVID-19 infection in East London: an observational cohort study BMJ Open 2021;11:e042140. doi: https://doi.org/10.1136/ bmjopen-2020-042140
- Smart A, Harrison E. The under-representation of minority ethnic groups in UK medical research. Ethn Health. 2017;22(1):65–82. https://doi.org/10. 1080/13557858.2016.1182126 (Epub 2016 May 13).
- Salway S, Holman D, Lee C, McGowan V, Ben-Shlomo Y, Saxena S, et al. Transforming the health system for the UK's multiethnic population. BMJ. 2020;368: m268. https://doi.org/10.1136/bmj.m268.
- Dawson S, Campbell SM, Giles SJ, Morris RL, Cheraghi-Sohi S. Black and minority ethnic group involvement in health and social care research: a systematic review. Health Expect. 2018;21:3–22. https://doi.org/10.1111/ hex.12597.
- 8. Health Research Authority https://www.hra.nhs.uk/planning-and-impro ving-research/best-practice/public-involvement/. Accessed 19 Oct 2021
- Papoulis SC, Callard F. 'A limpet on a ship': spatio-temporal dynamics of patient and public involvement in research. Health Expect. 2021;00:1–9.
- Ocloo J, Matthews R. From tokenism to empowerment: progressing patient and public involvement in healthcare improvement. BMJ Qual Saf. 2016;25(8):626–32.
- Russell J, Fudge N, Greenhalgh T. The impact of public involvement in health research: what are we measuring? Why are we measuring it? Should we stop measuring it? Res Involv Engagem. 2020;6:63. https://doi. org/10.1186/s40900-020-00239-w.
- Staley K. Changing what researchers "think and do": Is this how involvement impacts on research? Res All. 2017;1(1):158–67. https://doi.org/10. 18546/RFA.01.1.13.
- Staley K, Abbey-Vital I, Nolan C. The impact of involvement on researchers: a learning experience. Res Involv Engagem. 2017;3:20. https://doi.org/10.1186/s40900-017-0071-1.
- Scholz B, Bevan A. Toward more mindful reporting of patient and public involvement in healthcare. Res Involv Engagem. 2021;7:61. https://doi. org/10.1186/s40900-021-00308-8.

- NIHR Payment guidance for researchers and professionals. Version 1.0 April 2020 https://www.nihr.ac.uk/documents/payment-guidance-for-researchers-and-professionals/27392 Accessed 19th Oct 2021
- Taking Stock NİHR public involvement and engagement Version 1.0 June 2019 Accessed 19th Oct 2021
- Mulugeta B, Williamson S, Monks R, Hack T, Beaver K. Cancer through black eyes: the views of UK based black men towards cancer—a constructivist grounded theory study. Eur J Oncol Nurs. 2017;29:8–16. https:// doi.org/10.1016/j.ejon.2017.04.005.
- 18. Green G, Johns T. Exploring the relationship (and power dynamic) between researchers and public partners working together in applied health research teams. Front Sociol. 2019;4:20.
- 19. Pearce, C. The complexities of developing equal relationships in patient and public involvement in health research. Soc Theory Health (2020)

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- $\bullet\,$ thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

